

June 30, 2017

Dear Editor,

Title: Portal Vein Thrombosis Following Irreversible Electroporation for Locally Advanced Pancreatic Cancer

Author: Jun Jun Su, Ming Su, Kai Xu, Peng Fei Wang, Li Yan, Shi Chun Lu, Wan Qing Gu, and Yong Liang Chen

Name of Journal: *World Journal of Gastroenterology*

ESPS Manuscript NO: 34101

The manuscript has been improved according to the suggestions of reviewers:

The manuscript has been edited by a native English speaker.

According to the comments of the reviewers, we have revised our manuscript. The revised manuscript and the detailed responses to the comments of the one reviewer are attached.

Sincerely yours,

Jun Jun Su

Department of Hepatobiliary Surgery

Chinese People's Liberation Army General Hospital

Beijing 100853, China.

Email: chenyongl301@163.com

Telephone: +86-13601021099